Author’s response to reviews

Title: Prevalence of urinary colonization by ESbetaL Enterobacteriaceae among permanent catheterised inpatient in Italian LTCFs

Authors:

Luca Arnoldo (arnoldo.luca@libero.it)
Roberta Migliavacca (r.miglia@unipv.it)
Laura Regattin (laura.regattin@ass3.sanita.fvg.it)
Annibale Raglio (araglio@ospedaliriuniti.bergamo.it)
Elisabetta Nucleo (e.nucleo@unipv.it)
Melissa Spalla (melissa.spalla@unipv.it)
Francesca Vailati (fvailati@ospedaliriuniti.bergamo.it)
Laura Pagani (laura.pagani@unipv.it)
Silvio Brusaferro (brusaferro.silvio@aoud.sanita.fvg.it)

Version: 5 Date: 5 June 2012

Author’s response to reviews:

Dear editor,

I would like to submit the following article: “Prevalence of urinary colonization by ESBL Enterobacteriaceae among permanent catheterised inpatient in Italian LTCFs.” hoping it will be of interest to you.

In submitting this research article I warrant, on behalf of myself and my co-authors, that the article has not been formally published in any other peer-reviewed journal, is not under consideration by any other journal and does not infringe any existing copyright or any other third party rights. Moreover, no related papers have been published.

Besides no conflict of interest concern this study.